**Author’s response to reviews**

**Title:** The VICI-trial: High frequency oscillation versus conventional mechanical ventilation in newborns with congenital diaphragmatic hernia: An international multicenter randomized controlled trial

**Authors:**

- Lieke van den Hout (L.vandenhout@erasmusmc.nl)
- Dick Tibboel (D.tibboel@erasmusmc.nl)
- Sanne Vijfhuize (S.vijfhuize@erasmusmc.nl)
- Wim Hop (W.Hop@erasmusmc.nl)
- Irwin Reiss (I.reiss@erasmusmc.nl)

**Version:** 2  **Date:** 15 April 2011

**Author’s response to reviews:**

Dear editor,

Hereby I sent you a revised version of the manuscript. We stated that participating centres are still included and that ethical approval is still sought in one centre.

yours sincerely,

Lieke van den Hout